September, 2003

Information for Math 250, Introduction to Linear Algebra, Section 10

Lecturer

Yi-Zhi Huang Office and telephone number: Hill Center 332, 445-1314 (messages may be left at 445-2390). Office Hours: Monday and Wednsday 2:30 – 3:30 pm, other times by appointment, Hill Center 332.

E-mail address: yzhuang@math.rutgers.edu Webpage: http://www.math.rutgers.edu/~yzhuang/

Webpages

For Section 10: http://www.rci.rutgers.edu/~yzhuang/math/250-f-03.html For Math 250: http://www.math.rutgers.edu/courses/250/math250.html

Text

Spence, Insel & Friedberg: Elementary Linear Algebra: A Matrix Approach, Prentice-Hall.

Grading Scheme

Two midterms: 20% each Quizzes anbd homework problems: 30% Final exam: 30%

Homework will be assigned and collected but only selected ones will be graded. It is important to do the homework problems and hand them in regularly.

First Midterm Exam October 8, Wednsday, usual class time and place.

Second Midterm Exam November 12, Wednsday, usual class time and place.

Final Exam

December 17, Wednsday, 8 – 11 am, BE 003 (to be confirmed).